

# **Earth's Future**

## **RESEARCH ARTICLE**

10.1029/2025EF006188

#### **Key Points:**

- A non-stationary regional regression is used to estimate variability in annual elasticity estimates for the USA
- Annual streamflow elasticity to precipitation is more sensitive to interannual climate variability in dry regions
- Statistically significant temporal trends in elasticity exist in some regions but are generally small

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

B. J. Anderson, bailey.anderson@slf.ch

#### Citation:

Anderson, B. J., Slater, L. J., Rapson, J., Brunner, M. I., Dadson, S. J., Yin, J., & Buechel, M. (2025). Stationarity assumptions in streamflow sensitivity to precipitation may bias future projections. *Earth's Future*, *13*, e2025EF006188. https://doi.org/10.1029/2025EF006188

Received 26 FEB 2025 Accepted 22 JUN 2025

#### **Author Contributions:**

Conceptualization: Bailey J. Anderson, Louise J. Slater, Manuela I. Brunner Data curation: Bailey J. Anderson, Louise J. Slater, Jiabo Yin, Marcus Buechel Formal analysis: Bailey J. Anderson Funding acquisition: Bailey J. Anderson Investigation: Bailey J. Anderson, Louise J. Slater, Jessica Rapson, Marcus Buechel Project administration: Bailey J. Anderson

**Resources:** Bailey J. Anderson, Louise J. Slater

Software: Bailey J. Anderson Supervision: Louise J. Slater, Simon J. Dadson

© 2025. The Author(s). Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## Stationarity Assumptions in Streamflow Sensitivity to Precipitation May Bias Future Projections

Bailey J. Anderson<sup>1,2,3</sup> , Louise J. Slater<sup>1</sup>, Jessica Rapson<sup>4</sup>, Manuela I. Brunner<sup>2,3,5</sup>, Simon J. Dadson<sup>1,6</sup>, Jiabo Yin<sup>7</sup>, and Marcus Buechel<sup>1,8</sup>

<sup>1</sup>School of Geography and the Environment, University of Oxford, Oxford, UK, <sup>2</sup>WSL Institute for Snow and Avalanche Research SLF, Davos Dorf, Switzerland, <sup>3</sup>Climate Change, Extremes and Natural Hazards in Alpine Regions Research Center CERC, Davos Dorf, Switzerland, <sup>4</sup>Department of Statistics, University of Oxford, Oxford, UK, <sup>5</sup>Institute for Atmospheric and Climate Science, ETH Zurich, Zurich, Switzerland, <sup>6</sup>NERC Centre for Ecology and Hydrology, Wallingford, UK, <sup>7</sup>State Key Laboratory of Water Resources Engineering and Management, Wuhan University, Wuhan, P. R. China, <sup>8</sup>Guy Carpenter, London, UK

**Abstract** Streamflow elasticity to precipitation is a metric which is used to estimate how responsive rivers are to changes in precipitation. It is commonly used to anticipate future impacts of climate change on streamflow and is assumed to be constant in time, despite evidence that this relationship varies with climatological and landscape changes. To assess the need for a more flexible definition, we present a large-sample non-stationary regional regression approach to estimate long-term trends and variability in interannual streamflow elasticity to precipitation in the USA. We find that elasticity is highly variable in water-limited catchments year-to-year, indicating high sensitivity to climate variability in arid regions. Statistically significant long-term trends in elasticity exist in some regions, but trend magnitude is generally small. We demonstrate that a single average estimate of elasticity may be a poor indicator of streamflow sensitivity to climate change. Consideration of the variability of response is essential for elasticity to be a useful hydrologic signature.

**Plain Language Summary** Streamflow elasticity to precipitation is a simple metric used to estimate how much river flow changes in response to shifts in precipitation, on average, over long timescales. This understanding is important for predicting how climate change might affect water availability. Often researchers implicitly assume that this relationship stays the same over time, but in reality, it can change due to shifts in climate and landscape conditions. In our study, we analyzed data from many rivers and used a regional regression model to estimate long-term trends and year-to-year changes in streamflow response to precipitation. We found that in dry areas, river flow can be very sensitive to climate variations, meaning that changes in rainfall have a relatively large impact on water in rivers. Some regions exhibit long-term trends in the response of streamflow to rainfall, but these trends are generally small. Our findings suggest that using a single estimate for elasticity oversimplifies the problem. To better understand how rivers will respond to climate change, it is important to account for variations in how streamflow reacts to precipitation over time.

## 1. Introduction

Some of the most fundamental questions in the environmental sciences deal with how changes in one environmental variable influence another. This is especially true as we deal with anthropogenic climate and environmental changes. In hydrology, we are particularly concerned with how the environment influences the water cycle, and especially how climate change may be expressed through, and increase the threat of, hydrologic hazards (Douville et al., 2021). Understanding the sensitivity of the hydrologic cycle to changes in climatic factors is essential for continued water security.

Evidence suggests that climatological shifts might result in more extreme floods and droughts as well as elevated risks associated with water availability (Abbott et al., 2019). However, there is rarely a one-to-one relationship between climatological changes and changes in the volume of water in rivers (Ivancic & Shaw, 2015; Sharma et al., 2018). Instead, the rainfall-runoff relationship is controlled by the combined effects of the climatological and landscape properties of a river catchment, which modulate the partitioning of precipitation between storage, runoff, and evaporative processes. Water storage capacity, both short- and long-term, plays a key role in modulating how streamflow responds to precipitation—acting as a buffer, thereby dampening and delaying runoff (Anderson et al., 2024; Staudinger et al., 2017). Changes in water storage availability, as a result of landscape or





23284277, 2025, 7, Downloaded

10.1029/2025EF006188 by UK Centre For Ecology & Hydrology, Wiley Online Library on [11/07/2025].

. See the Terms

climatological shifts, may result in substantial variation in the rainfall-runoff relationship in the future (Anderson et al., 2022; Bassiouni et al., 2016; Blum et al., 2020; Davenport et al., 2020; Neri et al., 2019; Slater & Villarini, 2017; Vicente-Serrano et al., 2019). Regardless of the direction of change, if not considered, this issue could dramatically bias projections of future changes in water resources, with potentially serious implications for infrastructure, health, and quality of life.

Much like climate, river basins are complex natural systems, meaning that, with few exceptions, it is difficult to study changes in a controlled or experimental way. Instead, hydrologists often rely on modeling, statistical analysis, and proxies such as hydrologic signatures, to decipher physical processes and study the response of river basins to changes in climate, land cover, and water demand. Hydrologic signatures are widely used numerical metrics that describe the properties or characteristics of streamflow (McMillan, 2021). These metrics function as a set of tools which aid in understanding the underlying physical processes which control streamflow. They are frequently used to reveal weaknesses in hydrological models, for model calibration, and as tool to learn about hydrologic systems in empirical analyses (Addor et al., 2018; McMillan, 2021).

"Streamflow elasticity" is one such signature which describes how responsive streamflow is to precipitation or other environmental variables such as land cover change (Schaake, 1990) at the seasonal, annual, or multi-annual (Zhang et al., 2023) timescale. It is frequently used to estimate future climate change impacts on streamflow (Addor et al., 2018; Awasthi et al., 2024). Streamflow elasticity, henceforth referred to as "elasticity," is most often estimated as a single number calculated for a given period of record (e.g., decades) representing the average relationship between precipitation and streamflow over time.

While the *average* sensitivity of streamflow to changes in mean precipitation in a catchment or region is potentially useful information, the metric is limited in several ways. For instance, streamflow elasticity to precipitation is poorly predicted by climate indices (such as snow fraction, aridity, and precipitation seasonality) as well as the physical and landscape characteristics of a catchment (Addor et al., 2018). Physical reasons for spatial variability in elasticity can be difficult to pinpoint (Andréassian et al., 2016), suggesting that the metric is not well understood, and therefore may not be very informative. Further, some have recognized the limitations of elasticity regarding its ability to capture potential non-linearities in the precipitation-streamflow relationship (Fu et al., 2007; Tang et al., 2019).

Finally, typical elasticity estimates fail to acknowledge or address the potential for substantial variation in these estimates from year to year, or in response to persistent physical or climatological changes within a catchment over time (Bassiouni et al., 2016; Harman et al., 2011). This is problematic, given that a large body of research demonstrates that streamflow is influenced by, for example, changes in land cover (Anderson et al., 2022; Blum et al., 2020; Han et al., 2022; Slater & Villarini, 2017), antecedent moisture conditions (Bennett et al., 2018; Han et al., 2022; Wasko et al., 2020), and groundwater influxes (Berghuijs & Slater, 2023; Tague & Grant, 2009), among others. Thus, a stationary estimate of elasticity seems physically implausible.

Recent work has aimed to address the problem by incorporating a storage component, represented by a proxy such as baseflow or the residuals of the water balance closure, into elasticity estimation approaches (Cooper et al., 2018; Konapala & Mishra, 2016; Tang et al., 2020; Zhang et al., 2023). The inclusion of these parameters has led authors to claim that "elasticity changes little over time since it already represents the adaptation of catchment response to changes in its drivers" (Zhang et al., 2023). This approach relies on the assumption that the baseflow index fully captures the effect of water storage within the same year, as well as the effect of water storage brought forward from previous years. Baseflow is associated with storage contributions to streamflow, for groundwater and other stores, but it is unlikely to fully capture these storage components (Kalbus et al., 2006) or the variability which catchments experience on an interannual basis. Thus, it is possible that the inclusion of this component does not fully account for variation in streamflow response over time.

Broadly speaking, a large body of literature exists that investigates the ways in which the relationship between precipitation and streamflow response varies over time. For instance, as far back as Horton (1933), discussed the properties of soil, water content and how these relate to event runoff. Recently, research has investigated how streamflow response to precipitation might change, for example following prolonged drought (Fowler et al., 2022; Saft et al., 2016), and how changes in precipitation phase, for instance, can have an effect on response intensity and timing (Berghuijs et al., 2014). Given the long history of research into how what drives this relationship, it is surprising that streamflow elasticity is generally accepted as a stationary metric. To our knowledge,

Validation: Bailey J. Anderson, Louise J. Slater, Jessica Rapson, Manuela I. Brunner, Simon J. Dadson, Jiabo Yin Visualization: Bailey J. Anderson Writing – original draft: Bailey J. Anderson

Writing – review & editing: Bailey J. Anderson, Louise J. Slater, Jessica Rapson, Manuela I. Brunner, Simon J. Dadson, Jiabo Yin, Marcus Buechel

Wiley Online





**Figure 1.** Conceptual diagram. Shows "elasticity" as (a) a single average value representing streamflow response across time (black horizontal line), and (b) the hypothesized temporally variant values (black time-varying line). In (a), elasticity is constant, and streamflow response to the same amount of precipitation input is consistent. In (b) streamflow elasticity varies relative to, as an example, antecedent wetness in previous years, resulting in a different streamflow response to the same precipitation volume falling in different years.

only one publication has explicitly quantified variability in streamflow elasticity to precipitation over time. Tang et al. (2020) investigated the elasticity of streamflow to interannual climate variability directly using an analytical approach based on the Budyko hypothesis (Tang et al., 2020). The authors assessed how sensitive streamflow elasticity to precipitation and potential evaporation is on average from year to year in the United States. They accounted for the storage component of the water balance more explicitly, by estimating the difference in the residuals of the annual water balance closure. With respect to streamflow elasticity to precipitation, they argued that the coefficients of variation are "small" and thus that variability is not of great importance. However, the coefficients range as high as 0.6, which not particularly small. Further, the relatively large variability which they do find is generally in arid regions. These catchments were few and the attribution to irrigation posited by the authors is not empirically supported.

Therefore, there remains much more to be learned about the variable nature of hydrologic sensitivity to streamflow. In particular, to address this research gap, we assess how climate variability influences elasticity over time and in regions with different climatological norms. Figure 1 shows a conceptual representation of elasticity estimated using a typical approach in which it is represented as a single value over time (a), and a hypothetical representation of elasticity varying over time (b).

In this manuscript, we aim to quantify climate-driven variability in interannual streamflow elasticity and question the extent to which streamflow elasticity, as a hydrologic signature, may be biased by assumptions of temporal stationarity. More specifically, we address the following questions: "Is streamflow sensitivity to precipitation robust to interannual climate variability and well represented by a mean long-term estimate of elasticity?," "Do the regional interannual elasticity estimates exhibit long-term trends in elasticity which can be attributed to climate or anthropogenic drivers?." Capturing these aspects of nonstationarity in streamflow sensitivity carries implications for the use of elasticity as a hydrologic signature and for future streamflow projection. Additionally, it may provide evidence of the need to investigate variability in the response of similar metrics within the earth and environmental sciences. Our primary focus in this manuscript is on the median annual streamflow, and some additional results for low and high flows are included in the Supporting Information S1.

## 2. Materials and Methods

## 2.1. Methodological Justification

The approach used in this paper relies on a regional panel regression model to develop interannual elasticity estimates. To isolate climate effects and be able to capture interannual variability in the sensitivity of streamflow using observed data, the elasticities are calculated for climatologically similar regions. The use of the regions is essential for calculating interannual elasticities when using an observed, empirical approach, because it increases the degrees of freedom, allowing sufficient annual-scale data for estimations to be made with a reasonable degree of confidence. The regional approach also offers several advantages in that it allows us to capture the influence of climate, a characteristic which largely exhibits regional-scale behavior, on streamflow sensitivity to temporal changes in precipitation, while data pooling allows for more robust estimates of elasticity in each year. Within the regions, catchment-scale hydrological behavior may also be influenced by the physiographic characteristics of a catchment. We rely on the model design to isolate interannual and long-term variability which is the result of processes that occur at the regional scale, namely, climate, and to control for catchment characteristics (Anderson et al., 2022; Blum et al., 2020).

The panel regression approach which we use has been shown to adequately control for time invariant differences at the catchment scale and to facilitate the isolation of regional-scale effects (Anderson et al., 2022; Bassiouni et al., 2016; Blum et al., 2020; Steinschneider et al., 2013). Similar regional regression approaches have been popular in work related to predictions in ungauged basins (Cutore et al., 2007; Hailegeorgis & Alfredsen, 2017; Vogel et al., 1999). The primary disadvantage of the methodology is that the data aggregation used to strengthen statistical robustness also results in a reduction in the specificity of results; we estimate temporally variable, but spatially averaged elasticity values for each hydro-climatic region. This means that the results are best interpreted relative to one another (i.e., as a representation of change over time) rather than as individual elasticity estimates. While it is well known that elasticity varies substantially across space (Addor et al., 2018; Anderson et al., 2024; Sankarasubramanian et al., 2001), climatology has a more persistent spatial pattern. Thus, it is assumed that interannual variability in elasticity resulting from the regionalized model can be attributed to climate. In other words, the results can be interpreted as regional averages, and variability in elasticity as an approximation of the relative sensitivity experienced in each region, rather than precise estimates of elasticity at individual locations. Further, spatial variation in elasticity is frequently attributed to long-term climate, or at least described in terms of climate zones, and there is a strong connection between elasticity and the runoff ratio, lending further credibility to the approach (Ho et al., 2022).

## 2.2. Observed Hydrologic and Climate Data

The daily streamflow time series were obtained for the period 1981–2020 from the USGS website (https:// waterdata.usgs.gov/nwis) using the R package dataRetrieval (DeCicco et al., 2024) and were selected from the Geospatial Attributes of Gages for Evaluating Streamflow (GAGES II) catchments (Falcone, 2011). These catchments were filtered so that streamflow records used in the analysis had at least 30 years of 95% complete consecutive daily streamflow data between 1 January 1981 and 31 December 2021. Daily streamflow records were converted to water years, defined here as October through September, so that snowfall seasons occurred within the same water year. Since timeseries based on water years require data from the previous calendar year, water years 1981 and 2021 were excluded from the final analysis because they are incomplete. The 10th, 50th, and 90th percentiles of daily streamflow were calculated for each water year.

Ephemeral streams, defined as records including zero flow days were removed from the data set because we rely on a log-log linear regression approach for this analysis. This resulted in a total sample size of 2967 catchments. Separately, a subsample of 830 catchments meeting the same criteria but with minimal regulation (Anderson et al., 2022; Blum et al., 2020) was selected. We perform the same analysis with this subsample to confirm that dam storage had little influence on the overall detected relationships. Data used to estimate upstream dam storage was taken from GAGES II data set, as were watershed boundaries (Falcone, 2017). Upstream dam storage was calculated by dividing total upstream dam storage by annual average inflow (Falcone, 2011) resulting in a value with units in days. Stream gages with more than one day of upstream dam storage were eliminated (Blum et al., 2020; Hodgkins et al., 2019).



## **Earth's Future**

#### 10.1029/2025EF006188



**Figure 2.** Average interannual variability of elasticity and catchment attributes by region. (a) Map of hydro-climatologic regions used for grouping catchments in the model, colored according to the coefficient of variation (CV) of the interannual elasticity to precipitation. Numbers are the total number of catchments within each region in the study. (b–e) Box plots showing the distribution of the (b) aridity index (dimensionless), (c) average temperature (degrees Celsius), (d) the runoff ratio (dimensionless), and (e) the snow fraction (in percent of day/year) for individual catchments in each region. Center line is the median, boxes show the 25th and 75th percentiles, and whiskers represent the interquartile range multiplied by  $\pm 1.5$  from the 25th and 75th percentiles. Regions are ordered on the *x*-axis from least to most variable based on coefficient of variation and colored as in sub-plot a. Stars indicate a significant difference in the median of each variable of interest between the groups with the least variable (average CV < 10%) and most variable interannual elasticity (average CV > 30%): p < 0.001 (\*\*\*); p < 0.01 (\*\*); p < 0.05 (\*).

Gridded monthly precipitation and temperature (4 km resolution) were extracted from the Oregon State PRISM project data set (https://prism.nacse.org/recent/) using the R package prism (Edmund & Bell, 2015). We estimated average daily precipitation (mm/day) for each year from the monthly raster grid within each catchment boundary. We calculated average daily potential evaporation (mm/day) for each year in R using the Hamon equation (Hamon, 1963; Lu et al., 2007) with monthly temperature as previously described, and solar radiation estimated from latitude and Julian date (Equations S5–S7 in Supporting Information S1). The Hamon equation was used to retain consistency with the GAGES II data set and because this method has been shown to perform well relative to other approaches, despite its simple formulation (Lu et al., 2007). Annual values for climatological variables were calculated for water years, to coincide with streamflow data and snowfall seasons.

The Bukovsky regions (Bukovsky, 2011) were used to define the nine hydro-climatic regions used in this analysis (Figure 2a). These were selected because they offer a `middle ground' between climatological and hydrologically relevant regions (Brunner & Dougherty, 2022). We also tested the regionalization approach (Figure S1 in Supporting Information S1) presented in Knoben et al. (2018) which aims to address the shortcomings of

climatological regions for hydrological research and compares a classification approach based on hydrologic signatures to the Köppen-Geiger climate zones, in which they find that their index based approach is better. We replicated their methodology and found that the Bukovsky regions perform as well or better than the approach used in Knoben et al., 2018 (Figure S1 in Supporting Information S1 while maintaining a much higher degree of spatial contiguity, and therefore climatological similarity.

#### 2.3. Panel Model Setup

We estimate average interannual elasticity using a fixed effects panel regression model. These models resemble those which have been used for similar problems in hydrology (Anderson et al., 2022, 2024; Bassiouni et al., 2016; Blum et al., 2020). They allow for consideration of double indexed data, while controlling for time-invariant confounding variables at the catchment level, making them more robust than a typical cross sectional or time series regression model. The model is designed so that time-invariant confounding variables which exist at the catchment scale are controlled for by the fixed effects. Thus the captured variability in streamflow elasticity to precipitation is attributable to drivers which exist at the scale of the hydroclimatic regions. We assume that regional-scale changes predominantly relate to climate.

The regional average interannual elasticity estimates were calculated for three percentiles of streamflow: the 10th, 50th, and 90th using the following model:

$$\ln(Q_{it}) = \alpha_i + \sum_{r,y} \varepsilon_{P(r,y)} \cdot \ln(P_{it}) \cdot \mathbf{1}_{\text{region}_{i=r}} \cdot \mathbf{1}_{\text{year}_{r=y}} + \sum_r \varepsilon_{E(r)} \cdot \ln(E_{it}) \cdot \mathbf{1}_{\text{region}_{i=r}} + \sum_r \varepsilon_{BFI(r)} \cdot BFI_{it} \cdot \mathbf{1}_{\text{region}_{i=r}} + \eta_{it}$$

$$(1)$$

where  $\ln(Q_{it})$  represents the natural logarithm of the streamflow percentile (Q10, Q50, or Q90) for catchment *i* in year *t*. The percentile streamflow values are calculated annually, with Q10 representing low flows, Q50 median flows, and Q90 high flows. The term  $\alpha_i$  is a streamgage-specific intercept that accounts for time-invariant, catchment-specific confounding variables through the inclusion of individual fixed effects. The variable  $\ln(P_{it})$  is the natural logarithm of the annual daily mean precipitation averaged over each catchment *i* in year *t*,  $\ln(E_{it})$  is the natural logarithm of the annual daily mean potential evaporation for each catchment and year. BFI<sub>it</sub> denotes the annual baseflow index, calculated using the UK Institute of Hydrology's "smoothed minima" approach (Gustard et al., 1992) and a standard 5 days block size (Stoelzle et al., 2020) and the R package *lfstat* (Gauster et al., 2022).

Although the streamflow percentile response variable varies across models (Q10, Q50, Q90), the same set of predictors—namely precipitation, potential evaporation, and BFI—is used to estimate each. This allows us to evaluate the sensitivity of low, median, and high flows to changes in total annual precipitation and associated hydrometeorological drivers. The main text focuses on median flows (Q50), with additional results for Q10 and Q90 presented in the Supporting Information S1.

The interaction terms between precipitation, potential evaporation, BFI, and region indicators are used to allow these predictors to have region-specific effects. Specifically, the precipitation elasticity varies by both year and region and is captured by the coefficient  $\varepsilon_{P_{r,y}}$ . The coefficients for potential evaporation and BFI are assumed constant over time but vary across regions and are denoted  $\varepsilon_{E(r)}$  and  $\varepsilon_{BFI(r)}$  respectively. This modeling structure is equivalent to fitting separate regression models by region, however, it does influence the model adjusted  $\mathbb{R}^2$  and AIC. Therefore, we additionally calculated the models for the regions individually to estimate the model adjusted  $\mathbb{R}^2$  values.

The error term  $\eta_{it}$  represents the residual unexplained variation for catchment *i* in year *t*. BFI is included as a covariate due to its use in previous elasticity studies as a proxy for hydrologic storage. While the coefficient on BFI reflects a regionally averaged relationship between baseflow and streamflow, we acknowledge that its effect likely varies at the catchment scale. This is due to spatial heterogeneity in groundwater contributions, geology, and other physical catchment characteristics. Despite this, we include BFI to facilitate comparisons with related published models.



Equation 1 can be rearranged to solve for interannual precipitation elasticity in each region so that:

$$\varepsilon_{P_{\text{it}}} = \frac{\ln(Q_{\text{it}}) - \alpha_i - \varepsilon_{E(r)} \cdot \ln(E_{\text{it}}) - \varepsilon_{\text{BFI}(r)} \cdot \text{BFI}_{\text{it}} - \eta_{\text{it}}}{\ln(P_{\text{it}}) \cdot 1_{\text{region}_{i=r}} \cdot 1_{\text{year}_{r=v}}}$$
(2)

The model is estimated in *R* using the *plm* package (Croissant & Millo, 2008) and the "within" estimator is used to define fixed effects. The catchment level fixed effects control for time-invariant confounding variables at the catchment level, for example the physical characteristics of individual catchments, such as soil type and geology. By controlling for time-invariant confounders at the catchment level, we isolate changes in  $\varepsilon_{P(it)}$  which occur at the regional scale on a year-to-year basis, effectively limiting these effects to climatic changes. Thus, the interannual coefficients represent the effect of climate on the elasticity of streamflow to precipitation.

Autocorrelation in fixed effects panel models can lead to the underestimation of standard errors. We address this concern by clustering standard errors at the streamgage level (Anderson et al., 2022, 2024; Bertrand et al., 2004; Blum et al., 2020). A generalized R script for the application of these approaches is available in the Supporting Information S1 (Text S1).

#### 2.4. Model Comparison

We fit two additional forms of this model for comparison. The first is a fixed effects panel regression model which excludes the interaction term for "year" as below:

$$\ln(Q_{it}) = \alpha_i + \sum_r \varepsilon_{P(r)} \cdot \ln(P_{it}) \cdot \mathbf{1}_{\operatorname{region}_{i=r}} + \sum_r \varepsilon_{E(r)} \cdot \ln(E_{it}) \cdot \mathbf{1}_{\operatorname{region}_{i=r}} + \sum_r \varepsilon_{BFI(r)} \cdot BFI_{it} \cdot \mathbf{1}_{\operatorname{region}_{i=r}} + \eta_{it}$$
(3)

where all terms are the same as in Equation 1. This model estimates the average elasticity across the period of record without considering the effects of temporal change in precipitation on streamflow.

The second model form is a version of Equation 1 which excludes the annual baseflow index (BFI), as in Equation 4 below.

$$\ln(Q_{it}) = \alpha_i + \sum_{r,y} \varepsilon_{P(r,y)} \cdot \ln(P_{it}) \cdot \mathbf{1}_{\operatorname{region}_{i=r}} \cdot \mathbf{1}_{\operatorname{year}_{i=y}} + \sum_r \varepsilon_{E(r)} \cdot \ln(E_{it}) \cdot \mathbf{1}_{\operatorname{region}_{i=r}} + \eta_{it}$$
(4)

where all terms are the same as in Equation 1. This model is estimated to determine if the inclusion of BFI substantially reduces the effect of interannual variability on the model.

The results of these two models are compared with the results of the time variant model in Equation 1 using the Akaike information criterion (AIC; Equation S8 in Supporting Information S1), and indicate that the model which accommodates temporal variation in the precipitation elasticity of streamflow as well as annual BFI as in Equations 1 and 2 outperforms the long-term average model (Equation 3) and the temporally variant model which excludes BFI (Equation 4) for every flow percentile tested (Table S1 in Supporting Information S1).

Trends in the interannual elasticity estimates were calculated using a trend-free pre-whitened version (Bayazit & Önöz, 2007) of the widely applied Mann Kendall trend test (Hamed, 2008; Kendall, 1948; Mann, 1945). This test was applied using R package modifiedmk (Jassby et al., 2022).

The standardized precipitation index (SPI; Equation S1 in Supporting Information S1) and the standardized temperature anomaly (STA; Equation S2 in Supporting Information S1) are used to assess the relationship between streamflow and climate anomalies. These were calculated from annual regional average precipitation and temperature using a 12-month aggregation window, by taking the mean of the catchment averaged values in each region. The runoff ratio was calculated as the average total annual streamflow in mm divided by the average total annual precipitation (mm). Note that our primary focus in this manuscript is on the median annual streamflow (Q50). Additional results for low and high flows are included in the Supporting Information S1.

	Та	ble	1
--	----	-----	---

Mean Absolute and Relative Difference in the Regional Interannual Elasticity Estimates As Well As the Long-Term Trends in Elasticity

	a. Pacific northwest	b. East	c. South	d. Great lakes	e. Mountain west	f. Prairie	g. Pacific southwest	h. Great plains	i. Desert
Long-term estimate	1.42	1.74	2.22	1.41	1.14	2.66	1.34	1.35	0.63
Interannual range	1.29–1.55	1.26-1.92	1.77-2.43	0.95-1.57	0.79-1.24	1.41-2.87	0.23-1.36	0.24-2.03	-0.07-1.44
Mean relative diff.	4%	6%	6%	11%	13%	18%	31%	23%	48%
Mean absolute diff.	0.05	0.11	0.14	0.15	0.15	0.48	0.42	0.31	0.30

*Note.* Values are estimated using the whole period of record (39-year) and the full interannual range for median streamflow. "Mean absolute difference" is the average difference between streamflow elasticity estimated as a single value and the individual annual elasticity estimates in percentage points—meaning that they correspond to an absolute difference in elasticity. "Mean relative difference" is calculated by dividing the average difference in interannual elasticity from the long-term estimate and then normalizing by the long-term estimate (mean absolute diff. divided by long-term estimate).

## 3. Results

## 3.1. Justification of a Temporally Variable Model

We test and compare three different model formulations using the Akaike information Criterion (AIC; Equation S8 in Supporting Information S1). The results of this comparison indicate that the model which incorporates both temporal variability, and annual BFI in the precipitation elasticity of streamflow (Equations 1 and 2) outperforms the long-term average model (Equation 3) and the temporally variant model which excludes BFI (Equation 4) for every flow percentile tested (Table S1 in Supporting Information S1). These results indicate that allowing for temporal variability improves model performance and that the inclusion of a storage component (BFI) alone is insufficient to capture variability in its entirety (e.g., Equation 3).

## 3.2. Regional Scale Interannual Variability

We use the temporally variable model to estimate regional average long-term and interannual elasticity for nine hydro-climatic regions in the United States. Long-term elasticity estimates here are calculated using an approach akin to those typical to the discipline, as a single value representing the average across the complete timeseries. In this study, these are averages for an entire hydroclimatic region rather than a single catchment.

We find long-term elasticity estimates ranging from approximately 0.63 in the Desert region to 2.66 in the Prairie region (Table 1). These values are interpreted as the corresponding proportional change in the median annual streamflow for a 1% change in total annual precipitation. In other words, an elasticity of 0.63 suggests that median streamflow in the Desert region increases by about 6.3% when precipitation increases by 10%. Meanwhile, an elasticity of 2.66 indicates that precipitation changes are magnified in streamflow, pointing to a streamflow increase of approximately 26.6% if precipitation increases by 10%.

However, the interannual elasticity can deviate greatly from these long-term elasticity estimates (Table 1). For instance, the mean absolute difference between the long-term elasticity estimates and the interannual estimates for median streamflow in the Prairie region is approximately 0.5. This means that on average across the 39-year period, the elasticity in each year deviates from the long-term mean by half a percentage point. Thus, elasticity estimates in each year range between about 1.4 and 2.9. This corresponds to a mean relative difference of 18% between elasticity estimated as a single long-term value and the mean interannual estimate (Table 1). In other words, a 10% change in precipitation could result in anywhere from a 14%–29% change in streamflow, assuming changes magnify linearly. In other words, while the coefficient of variation (Table 2) in the Prairie region is relatively small (0.17), these values represent a substantial difference in the context of streamflow projection. Similarly, in the Desert region, where the long-term elasticity is 0.63, the interannual estimates range from approximately 0 to 1.4, meaning that there could be anywhere from no response to a response in which precipitation is magnified in flow.

The range of interannual elasticity estimates varies substantially between regions, with some regions, like the Desert, Great Plains, and Pacific Southwest, showing large fluctuations in responsiveness, and others, like the Pacific Northwest and the East showing limited variability. These differences are clear and statistically significant between the nine Bukovsky regions (Bukovsky, 2011) (Figure 2).

#### Table 2

Results of the Mann Kendall Trend Test on Interannual Elasticity and the Coefficient of Variation of Interannual Elasticity for the 50th Percentile of Flow

		a. Pacific northwest	b. East	c. South	d. Great lakes	e. Mountain west	f. Prairie	g. Pacific southwest	h. Great plains	i. Desert
Mann Kendall trend test	Total change	0.06	-0.04	-0.28	-0.07	0.04	0.05	0.33	-0.37	-0.58
	p value	0.145	0.669	0.003	0.513	0.706	0.88	0.013	0.066	0.01
Coefficient of variation		0.04	0.08	0.08	0.12	0.13	0.17	0.27	0.33	0.58
Model adjusted R <sup>2</sup>		0.62	0.58	0.57	0.5	0.32	0.58	0.51	0.31	0.36

*Note.* Statistically significant trends (p < 0.05) are presented in bold font. Total change is the Sen's slope rate of change multiplied by 39 years. This is a unitless value representing the linearly estimated difference in streamflow elasticity to precipitation in water year 1982 versus water year 2020. The coefficient of variation in each region over each time period is presented as a percentage. Model performance is represented by the model adjusted  $R^2$  presented for each region.

The regions with the highest variability tend to be more water-limited (higher aridity index and lower runoff ratio) and receive the majority of precipitation as rain rather than snow, although not exclusively (Figures 2b–2e). The most variable regions also have the lowest runoff ratios, in line with existing literature (Chiew et al., 2006). Meanwhile the regions which experience the least interannual variability in streamflow elasticity to precipitation (e.g., the East and South) are also warm and rain-dominated, but climatic conditions in these regions are far more humid on average (Figure 2). The precise estimates of elasticity in each year are included in Figure 3, demonstrating how the range of hydrological responses can vary widely from year to year.

#### 3.3. Long-Term Trends in Interannual Elasticity

In addition to interannual variability due to climate fluctuations, it is possible that elasticity could exhibit long term changes due to shifts in typical climate patterns. For example, recent evidence has suggested that prolonged drought can have a lasting influence on streamflow sensitivity to precipitation (Fowler et al., 2022; Saft et al., 2016) and changes in precipitation phase, for instance, can have an effect on response intensity and timing (Berghuijs et al., 2014). Therefore, we investigate the influence of climate on average streamflow elasticity to precipitation over time by examining the presence of trends in the interannual estimates.

Results show that strongly statistically significant monotonic trends (Mann-Kendall p < 0.01) are present in the Desert, South, and Pacific Southwest regions (Figure 3; Table 2; c, g, i). This indicates that elasticity could be changing, on average, in these regions over time. Here, in addition to interannual variability, the relationship between streamflow and precipitation may be non-stationary even as an average. The Desert and South regions show negative trends, indicating that streamflow has become less responsive to precipitation change over the study period, while elasticity in the Pacific Southwest has increased. For the most part, however, the magnitude of the trends is small (less than 0.6; Table 2). Therefore, while statistically significant, these trends may not represent a substantial change in average streamflow sensitivity to precipitation over the 39-year study period. Regardless, this demonstrates the possibility of long-term trends in elasticity, which may influence the appropriateness of the metric for future streamflow projection.

We reran the analysis with a sub-sample of data (Table S3 in Supporting Information S1) from which sites with substantial dam storage had been removed. Trends in the South and the Desert region (c, i) persisted in this smaller sub-sample. These results suggest that long-term trends in these regions may be due to climate or land cover changes, rather than regulation, with the caveat that this change resulted in substantially reduced sample sizes (Table S3 in Supporting Information S1) and therefore confidence in the estimates. Interannual variability remained relatively consistent in the models resulting from this subsample.

#### 3.4. Comparison of Interannual Variability to Climate

To assess how year-to-year climate variability influences streamflow sensitivity to precipitation, we compare region-scale annual standardized anomalies of precipitation and temperature to interannual elasticity. This simplified approach does not pretend to capture the complete effects of changes in antecedent conditions or memory effects on hydrologic response but is representative of a general climatological pattern.



## **Earth's Future**



**Figure 3.** Interannual variability and long-term trends in interannual elasticity estimates for the 50th percentile of streamflow by hydro-climatologic region. Linear trend lines are presented in black on panels (a–i), with significance test results in Table 2. Subplots (a–i) show individual regions, and subplot k. shows the distributions of elasticity for each region as boxplots, where the upper and lower hinges are the 25th and 75th quantiles, and the whiskers include all values  $\pm 1.5$  times the interquartile range. The boxplots in sub-plots (j, k) show the distribution of elasticity estimates per region across all years. The box plots in sub-plot k are normalized by the median that the range of values can be easily compared without considering the magnitude.

Comparing the interannual elasticity estimates to contemporary and lagged climatic variables reveals that the regional standardized precipitation index (SPI) is highly (R > 0.5 and p < 0.05) or moderately (0.5 > R > 0.3 and p < 0.05) positively correlated with interannual variability in elasticity in the previous, present, or both years in a majority of regions (89% for median flows). The five regions with the most variable interannual elasticity (e–i) were correlated with SPI in the same year or first lag for all examined flow percentiles (Figures 4e–4i). In the five regions with the most variable interannual elasticity (e–i in Figure 4), we show correlation with SPI values.





**Figure 4.** Pearson correlation between streamflow elasticity to precipitation and the regional average SPI. Shown for the same (lag\_0), the previous (lag\_1), or two water years prior (lag\_2) for the 50th percentile of streamflow. Significance is rounded to two digits and indicated by stars where p < 0.1 is represented by \*, p < 0.05 is represented by \*\* and p < 0.01 is represented by \*\*\*. The linear trend line is shown in black and the one-to-one trend line is shown in gray. SPI and Elasticity are unitless metrics and elasticity is normalized by the median to facilitate comparison between the regions.

Streamflow responsiveness depends on conditions in previous years; namely streamflow is more responsive to average annual precipitation if that year, the previous year, or both are wetter than average, while the opposite is true for drier years.

SPI is most frequently correlated with elasticity in the first lagged water year. The correlation between delayed SPI and elasticity is likely due in part to the seasonality of precipitation and streamflow (Table S2 in Supporting Information S1). For example, the low flow season for most catchments in the desert region is summer, thus occurring within the same water year as the high precipitation season (typically winter or summer). In many catchments in the Pacific Southwest region, the low flow season is autumn while the largest proportion of precipitation falls in winter which occurs after autumn in the water year (October–September). Thus, it follows that antecedent precipitation in the previous year could have a larger effect on low flows than precipitation events in the same year. Further, while the October through September water year is widely used in the United States, it may not correctly capture the complete hydrologic cycle in all locations.

Baseflow, the proportion of streamflow originating from delayed sources, also represents more than 10% of median streamflow in the Pacific Southwest, Mountain West, Pacific Northwest, East, and Great Lakes regions (a, b, d, e, g), based on the regional median ratio of annual average daily baseflow to the annual median streamflow (Table S4 in Supporting Information S1). This contribution may explain some of the pattern of delayed responses in median streamflow. The proportion is even larger in catchments without substantial dam storage (Table S4 in Supporting Information S1). Precipitation elasticity of high streamflow magnitude shows fewer correlations with SPI overall. Results indicate a correlation only in the five most variable regions.

Correlations between interannual elasticity and temperature, characterized using the regional standardized temperature anomaly (STA; Figure 5), are weaker and less consistent overall than those with SPI. However, STA is moderately negatively correlated (p < 0.05) with elasticity in the same year for median flows in the most arid regions, the Desert and Great Plains (h, i). At the lag-1 timestep (previous water year), STA is negatively correlated with interannual elasticity in the Great Plains, Prairie, and Great Lakes regions (d, f, h) for median flows. These results indicate that increased temperatures exert some negative influence on



**Figure 5.** Pearson correlation between streamflow elasticity to precipitation and the regional average STA. Shown for the same (lag\_0), the previous (lag\_1), or two water years prior (lag\_2) for the 50th percentile of streamflow. Significance is rounded to two digits and indicated by stars where p < 0.1 is represented by \*, p < 0.05 is represented by \*\* and p < 0.01 is represented by \*\*\*. The linear trend line is shown in black, and the negative one-to-one trend line is shown in gray. STA and Elasticity are unitless metrics and elasticity is normalized by the median to facilitate comparison between the regions.

streamflow elasticity on a year-to-year basis, likely as a result of increased potential evaporation and drying soils (Sharma et al., 2018), or in response to a reduction in rain days with increased temperatures (Wasko & Nathan, 2019).

## 4. Discussion

Building on our foundational hypothesis that *streamflow elasticity is unlikely to be stationary in time*, the objective of this paper is to examine the extent to which an average estimate of streamflow response to precipitation at the annual time scale, as characterized by elasticity, may fail to capture changes in hydrologic response over time. Our results demonstrate that elasticity does, in fact, vary over time at a regional scale. This variation occurs over multiple timescales—both interannually and, in some regions, as a long-term average—over the 39-year study period.

We find similar spatial patterns to previous research on interannual streamflow elasticity (Tang et al., 2020). For example, large variability generally occurs in more arid regions. However, in contrast to Tang et al. (2020) we do not consider interannual elasticity to precipitation to be small. Interannual elasticity estimates in some regions deviate widely from the long-term average estimate. Coefficients of variation show a mean relative difference in response to annual precipitation of over 20% in some regions, and up to almost 50% in the most extreme case (Desert). Future streamflow projections that assume stationary elasticity could be substantially over- or underestimated, and elasticity estimates may poorly describe hydrologic behavior if used as a regime descriptor. For instance, a projected median streamflow which is 50% higher than is realistic, would result in failure to capture potentially severe drought conditions, and may result in severe overestimation of groundwater recharge and availability. These concerns are further demonstrated in the results for low and high flows as presented in the Supporting Information S1 (Table S5 and S6).

Our results show that dry catchments are more susceptible to fluctuations in precipitation than catchments in humid regions. These findings generally align with prior literature regarding the variability of flow in arid and semi-arid catchments (Farquharson et al., 1992) and the importance of antecedent moisture conditions for generating flow in these regions (Ivancic & Shaw, 2015). For example, a recent analysis presented the concept of

23284277, 2025, 7, Download

an "antecedent effect ratio" (AER) which quantifies the difference in catchment response to heavy precipitation dependent on antecedent moisture conditions (Bennett et al., 2018). They found that more arid catchments had high AER values, indicating that antecedent conditions were more important for flood flow generation in drier catchments than wetter ones. The authors speculated that the difference in response was related to the range of antecedent moisture conditions in drier catchments (i.e., arid catchments become much drier than humid ones, so relative moisture has a larger impact) (Bennett et al., 2018). In other words, as we demonstrate, antecedent precipitation conditions have a larger influence on streamflow response to precipitation in these regions because the role of hydrologic memory is more clearly expressed (Berghuijs et al., 2025; de Lavenne et al., 2022).

Another recent study demonstrated that the water balance in arid regions is dominated by extreme rainfall events (Dykman et al., 2023), thus streamflow response is susceptible to shorter term storage components, that is precipitation in the recent past, rather than contributions from long-delayed sources. Rivers in the Great Plains, Prairie, and Desert regions of our analysis likely have a weaker seasonality and flows are dominated by the occurrence of short precipitation events such as storms (Brunner et al., 2020). The fast and slow flow thresholds, two parameters that indicate the amount of precipitation required to generate storm runoff and baseflow respectively, are likely to consume a large proportion of annual precipitation in arid catchments (Harman et al., 2011; Sivapalan et al., 2011). In other words, streamflow which is driven predominantly by weather events, rather than longer-term storage, may be more sensitive at short timescales to both antecedent and present climate conditions. It is therefore unsurprising that drier regions experience larger shifts in elasticity on an interannual basis, a fact which is further supported in some previous research.

Previous research has also suggested that drier regions experience higher, and more spatially variable elasticity (Harman et al., 2011; Sankarasubramanian et al., 2001). Many of these studies have relied on the MOPEX data set, which is geographically limited in the especially arid Desert and southern portion of the Great Plains. Contrary to the results of these studies, we find relatively low long-term average elasticities in the arid Desert region. However, highly variable interannual elasticities here indicate that previously identified high spatial variability in dry regions may also be reflected in temporal variability, and years which are highly responsive will be interspersed with those that are not. Streamflow sensitivity may be better represented in these cases by an index which takes into consideration the importance of individual precipitation events on longer-term streamflow. For instance, it is possible that some version of an event dependence index, for example similar to the event flow distributions assessed in Dykman et al. (2023), may provide a suitable proxy for variability in elasticity over time. This concept could be explored in future research.

We detect some significant long-term trends in the regional interannual elasticity estimates, indicating that overall streamflow sensitivity to precipitation may be changing in some regions and for some parts of the flow distribution. Previous research has indicated that elasticity is correlated with the Horton index, a water-balance-based indicator of catchment humidity which relates evaporation to catchment wetting (precipitation-surface runoff) (Harman et al., 2011). Shifts in this index at long timescales may cause changes in catchment wetting and vaporization thresholds, eventually leading to changes in elasticity as a new climatic equilibrium is achieved (Harman et al., 2011). Further, it is possible that prolonged drought conditions may influence the long-term relationship between precipitation and streamflow (Fowler et al., 2022; Saft et al., 2015), and it is the case that significant increasing trends in drought (Ficklin et al., 2015), as well as large scale decreases in terrestrial water storage (Slater & Villarini, 2016) have been detected in the Desert region and in large parts of the South. These patterns may carry some responsibility for the decreasing trends in elasticity in these regions, providing another avenue for future research.

We find that long-term trends in elasticity due to climate may be occurring in some places, but that the geographic extent and magnitude of these changes is, so far, limited. The statistically significant trends which we detect are small in magnitude, showing a total absolute estimated linear change over the 39-year period of between 0.28 and 0.58. Due to the regional design of our study, we focus here on climate impacts, however, other anthropogenic activities such as land cover changes are also likely to influence elasticity at smaller spatial scales—potentially outsizing the average climate-driven patterns we find here in some places (Auerswald et al., 2025).

Long-term trends in several regions appear to be explained by dam regulation, as removal of these catchments from the data sample eliminated trends in several regions. In these cases, it may be reasonable to assume that long-term trends are due to management of hydrological resources rather than changes in climatology. Regardless of the magnitude and extent of long-term trends in elasticity, we demonstrate here that they are possible, suggesting



that changes in climatology over the long term could lead to differences in the typical streamflow response to precipitation.

#### 4.1. Limitations

We focus on a first-order precipitation elasticity of streamflow, and do not consider the explicit interactions and subsequent non-linear nature of streamflow elasticity as precipitation and temperature (potential evaporation) vary simultaneously (Fu et al., 2007). The elasticity estimates presented in the study represent within-group averages for entire regions, and therefore are not likely accurate estimates of streamflow elasticity at individual locations. The fixed effects in the panel regression model control for time-invariant confounders at the catchment scale (ex. topography), and thus it is assumed that the interannual and long-term variability detected here is the result of processes occurring at the regional scale, principally climate.

These results can be interpreted as regional averages, and variability in elasticity as an approximation of the relative sensitivity experienced in each region. The variations in elasticity which are presented in this study are robust at the regional level, with some caveats. The sample sizes in each region influence the statistical significance of the annual estimates. This is most relevant to the estimates for the Desert region which contains fewer catchments than the others, the implications of which are visible in the breadth of the associated confidence intervals (Figure 3). Regardless, the majority ( $\sim$ 80%) of interannual elasticity estimates are statistically significant (p < 0.05) in the Desert region.

Further, the model adjusted  $R^2$  is generally low to moderate (Table 2), especially for low flows, while the model adjusted  $R^2$  for high flows shows better performance (Table S6 in Supporting Information S1). Low model adjusted  $R^2$  for low flows is expected because this section of the annual streamflow hydrograph is typically by delayed sources and is difficult to predict especially with a simple model formulation. The models used here were designed to capture the effect of temporally variant factors and were not focused on prediction but rather explanation. Given the careful construction of the model, this highlights the importance of delayed sources of runoff in explaining low flow behavior, while still capturing the role of precipitation in driving this streamflow. It is worth noting that the sample size within each region influences the precision and accuracy of regression coefficient estimation in the panel regression model. As noted in Figure 2a, the Desert region contains substantially fewer catchments than other regions in the analysis, which may partially explain the relatively wider confidence intervals and reduced significance of interannual elasticity.

#### 5. Conclusions

In this paper, we have used a series of straightforward regional regression models to assess the degree to which streamflow elasticity to precipitation might vary over time. The model design isolates the effects of regional-scale processes which vary in time and helps control for processes at the catchment scale.

We have demonstrated that elasticity can sometimes vary substantially from year to year. We have also shown that this variation corresponds, to a high degree, to climate conditions in the previous year(s). Further, the difference in model performance indicates that inclusion of storage processes via a baseflow index metric does not adequately correct this shortcoming, as has been assumed in some previous work. Interannual variability is largest in dry regions, highlighting the importance of hydrologic memory for determining streamflow response. Lastly, we have shown that, while small, statistically significant long-term trends in elasticity are present in some regions. This suggests that climate change could, over time, shift catchment response to precipitation, further complicating the predictability of future streamflow.

Our results carry several implications. First, in catchments which experience high degrees of variability in streamflow elasticity to precipitation, future streamflow projections which rely on a single average estimate of elasticity are likely to over- or under-estimate flow. Biases could be as high as 50% in some cases, according to our results. Second, when elasticity is used as a hydrologic signature, for example for catchment classification, process exploration, or model calibration, a long-term average value may be a poor proxy for streamflow responsiveness. For this reason, it may not be a useful metric, especially in dry regions. Finally, if for some reason, a single long-term average estimate of elasticity is sought, our results indicate that, in catchments which experience high variability or long-term trends in elasticity, the time period used to estimate this metric could have a substantial effect on the resulting value. In other words, a short period of record may distort elasticity estimates in

dry regions, while in regions where interannual variability in elasticity is marginal, a short period of record may still yield a reasonable representative estimate.

Thus, our results, combined with those presented in previous work (Addor et al., 2018), suggest that as it is currently estimated and understood, elasticity is a poorly informative hydrologic signature. So long as elasticity is poorly explainable by climate and catchment characteristics, shows limited predictability in space, as indicated in previous work (Addor et al., 2018), and is sometimes highly temporally variable, as demonstrated here, its usefulness may be quite restricted. Future research should investigate methods for estimation of the degree of variability in elasticity at individual catchments, or more generally, other approaches to understand rainfall-runoff responses (Kirchner, 2024).

Improving knowledge of why elasticity varies, both spatially and temporally, is an important step toward understanding how, and if, elasticity can be used to project streamflow changes into the future, or to validate simulations. These questions lend themselves to larger considerations which are not addressed here regarding how to manage hydrological non-stationarity, and the lengths of time series necessary to adequately estimate streamflow sensitivity from the observed record.

## **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

## **Data Availability Statement**

The calculated data used to produce the main results (Anderson, 2023) is available here: https://doi.org/10.5281/ zenodo.8370039. All original data used in this analysis is publicly available from the following webpages: total upstream dam storage, average annual runoff, drainage area, mean catchment elevation, latitude and average catchment slope are available through GAGES II (Falcone, 2011) at https://doi.org/10.3133/70046617 watershed boundaries are also available through GAGES II (Falcone, 2017) at https://doi.org/10.5066/F7HQ3XS4; climate data is available from PRISM at https://www.prism.oregonstate.edu/recent/ and can be downloaded using the prism R package (Edmund & Bell, 2015). Streamflow data can be downloaded from the National Water Information System (NWIS) using the R package dataRetrieval (DeCicco et al., 2024). Bukovsky regions are accessible through the North American Regional Climate Change Assessment Program at https://www.narccap. ucar.edu/data/access.html.

## References

- Abbott, B. W., Bishop, K., Zarnetske, J. P., Minaudo, C., Chapin, F. S., Krause, S., et al. (2019). Human domination of the global water cycle absent from depictions and perceptions. *Nature Geoscience*, 12(7), 533–540. Article 7. https://doi.org/10.1038/s41561-019-0374-y
- Addor, N., Nearing, G., Prieto, C., Newman, A. J., Le Vine, N., & Clark, M. P. (2018). A ranking of hydrological signatures based on their predictability in space. *Water Resources Research*, 54(11), 8792–8812. https://doi.org/10.1029/2018WR022606
- Anderson, B. (2023). Elasticity\_variability: Initial release to zenodo [Dataset]. https://doi.org/10.5281/zenodo.8370039
  - Anderson, B. J., Slater, L. J., Dadson, S. J., Blum, A. G., & Prosdocimi, I. (2022). Statistical attribution of the influence of urban and tree cover change on streamflow: A comparison of large sample statistical approaches. *Water Resources Research*, 58(5), e2021WR030742. https://doi. org/10.1029/2021WR030742
  - Anderson, B. J., Brunner, M. I., Slater, L. J., & Dadson, S. J. (2024). Elasticity curves describe streamflow sensitivity to precipitation across the entire flow distribution. *Hydrology and Earth System Sciences*, 28(7), 1567–1583. https://doi.org/10.5194/hess-28-1567-2024
  - Andréassian, V., Coron, L., Lerat, J., & Le Moine, N. (2016). Climate elasticity of streamflow revisited—An elasticity index based on long-term hydrometeorological records. *Hydrology and Earth System Sciences*, 20(11), 4503–4524. https://doi.org/10.5194/hess-20-4503-2016
  - Auerswald, K., Geist, J., Quinton, J. N., & Fiener, P. (2025). HESS opinions: Floods and droughts—Are land use, soil management, and landscape hydrology more significant drivers than increasing CO<sub>2</sub>? *Hydrology and Earth System Sciences*, 29(9), 2185–2200. https://doi.org/10.5194/ hess-29-2185-2025
  - Awasthi, C., Vogel, R. M., & Sankarasubramanian, A. (2024). Regionalization of climate elasticity preserves Dooge's complementary relationship. Water Resources Research, 60(10), e2023WR036606. https://doi.org/10.1029/2023WR036606

Bassiouni, M., Vogel, R. M., & Archfield, S. A. (2016). Panel regressions to estimate low-flow response to rainfall variability in ungaged basins. Water Resources Research, 52(12), 9470–9494. https://doi.org/10.1002/2016WR018718

- Bayazit, M., & Önöz, B. (2007). To prewhiten or not to prewhiten in trend analysis? *Hydrological Sciences Journal*, 52(4), 611–624. https://doi. org/10.1623/hysj.52.4.611
   Bennett, B., Leonard, M., Deng, Y., & Westra, S. (2018). An empirical investigation into the effect of antecedent precipitation on flood volume.
- Journal of Hydrology, 567, 435–445. https://doi.org/10.1016/j.jhydrol.2018.10.025

Berghuijs, W. R., & Slater, L. J. (2023). Groundwater shapes North American river floods. *Environmental Research Letters*, 18(3), 034043. https://doi.org/10.1088/1748-9326/acbecc

Berghuijs, W. R., Woods, R. A., Anderson, B. J., Hemshorn de Sánchez, A. L., & Hrachowitz, M. (2025). Annual memory in the terrestrial water cycle. Hydrology and Earth System Sciences, 29(5), 1319–1333. https://doi.org/10.5194/hess-29-1319-2025

J.Y. acknowledges support from the National Natural Science Foundation of China (Grants 52441902; 52361145864; W2421111). LJS is supported by UKRI (MR/V022008/1). BJA acknowledges funding from the Clarendon Scholarship and Hertford College Oxford.

- Berghuijs, W. R., Woods, R. A., & Hrachowitz, M. (2014). A precipitation shift from snow towards rain leads to a decrease in streamflow. *Nature Climate Change*, 4(7), 583–586. Article 7. https://doi.org/10.1038/nclimate2246
- Bertrand, M., Duflo, E., & Mullainathan, S. (2004). How much should we trust differences-in-differences estimates? *Quarterly Journal of Economics*, 119(1), 249–275. https://doi.org/10.1162/003355304772839588
- Blum, A. G., Ferraro, P. J., Archfield, S. A., & Ryberg, K. R. (2020). Causal effect of impervious cover on annual flood magnitude for the United States. *Geophysical Research Letters*, 47(5), e2019GL086480. https://doi.org/10.1029/2019GL086480
- Brunner, M. I., & Dougherty, E. M. (2022). Varying importance of storm types and antecedent conditions for local and regional floods. Water Resources Research, 58(12), e2022WR033249. https://doi.org/10.1029/2022WR033249
- Brunner, M. I., Melsen, L. A., Newman, A. J., Wood, A. W., & Clark, M. P. (2020). Future streamflow regime changes in the United States: Assessment using functional classification. *Hydrology and Earth System Sciences*, 24(8), 3951–3966. https://doi.org/10.5194/hess-24-3951-2020
- Bukovsky, M. S. (2011). Masks for the Bukovsky regionalization of North America [Dataset]. Regional Integrated Sciences Collective. http://www.narccap.ucar.edu/contrib/bukovsky/
- Chiew, F., Peel, M., McMahon, T., & Siriwardena, L. (2006). Precipitation elasticity of streamflow in catchments across the world. Climate Variability and Change—Hydrological Impacts (Proceedings of the Fifth FRIEND World Conference Held at Havana, Cuba, November 2006) (Vol. 308, pp. 256–262).
- Cooper, M. G., Schaperow, J. R., Cooley, S. W., Alam, S., Smith, L. C., & Lettenmaier, D. P. (2018). Climate elasticity of low flows in the maritime Western U.S. Mountains. *Water Resources Research*, 54(8), 5602–5619. https://doi.org/10.1029/2018WR022816
- Croissant, Y., & Millo, G. (2008). Panel data econometrics in R: The plm package. Journal of Statistical Software, 27(2). https://doi.org/10.18637/jss.v027.i02
- Cutore, P., Cristaudo, G., Campisano, A., Modica, C., Cancelliere, A., & Rossi, G. (2007). Regional models for the estimation of streamflow series in ungauged basins. *Water Resources Management*, 21(5), 789–800. https://doi.org/10.1007/s11269-006-9110-7
- Davenport, F. V., Herrera-Estrada, J. E., Burke, M., & Diffenbaugh, N. S. (2020). Flood size increases nonlinearly across the Western United States in response to lower snow-precipitation ratios. *Water Resources Research*, 56(1), e2019WR025571. https://doi.org/10.1029/ 2019WR025571
- DeCicco, L., Hirsch, R., Lorenz, D., Watkins, D., & Johnson, M. (2024). dataRetrieval: Retrieval functions for USGS and EPA hydrologic and water quality data (version 2.7.15) [R Software]. U.S. Geological Survey. https://doi.org/10.5066/P9X4L3GE
- de Lavenne, A., Andréassian, V., Crochemore, L., Lindström, G., & Arheimer, B. (2022). Quantifying multi-year hydrological memory with catchment forgetting curves. *Hydrology and Earth System Sciences*, 26(10), 2715–2732. https://doi.org/10.5194/hess-26-2715-2022
- Douville, H., Raghavan, K., Renwick, J. A., Allan, R. P., Arias, P. A., Barlow, M., et al. (2021). Water cycle changes. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 1055–1210). Cambridge University Press. https://doi.org/10.1017/9781009157896.001
- Dykman, C., Sharma, A., Wasko, C., & Nathan, R. (2023). Can annual streamflow volumes be characterized by flood events alone? *Journal of Hydrology*, 617, 128884. https://doi.org/10.1016/j.jhydrol.2022.128884
- Edmund, H., & Bell, K. (2015). prism: Access data from the Oregon state prism climate project (version 0.0.6) [R Software]. https://doi.org/10. 5281/zenodo.33663
- Falcone, J. A. (2011). GAGES-II: Geospatial attributes of gages for evaluating streamflow [Dataset]. In GAGES-II: Geospatial Attributes of Gages for Evaluating Streamflow. U.S. Geological Survey. https://doi.org/10.3133/70046617
- Falcone, J. A. (2017). U.S. Geological Survey GAGES-II time series data from consistent sources of land use, water use, agriculture, timber activities, dam removals, and other historical anthropogenic influences [Dataset]. U.S. Geological Survey. https://doi.org/10.5066/F7HQ3XS4
- Farquharson, F. A. K., Meigh, J. R., & Sutcliffe, J. V. (1992). Regional flood frequency analysis in arid and semi-arid areas. *Journal of Hydrology*, 138(3–4), 487–501. https://doi.org/10.1016/0022-1694(92)90132-F
- Ficklin, D. L., Maxwell, J. T., Letsinger, S. L., & Gholizadeh, H. (2015). A climatic deconstruction of recent drought trends in the United States. Environmental Research Letters, 10(4), 044009. https://doi.org/10.1088/1748-9326/10/4/044009
- Fowler, K., Peel, M., Saft, M., Peterson, T. J., Western, A., Band, L., et al. (2022). Explaining changes in rainfall–runoff relationships during and after Australia's Millennium drought: A community perspective. *Hydrology and Earth System Sciences*, 26(23), 6073–6120. https://doi.org/10. 5194/hess-26-6073-2022
- Fu, G., Charles, S. P., & Chiew, F. H. S. (2007). A two-parameter climate elasticity of streamflow index to assess climate change effects on annual streamflow. Water Resources Research, 43(11). https://doi.org/10.1029/2007WR005890
- Gauster, T., Laaha, G., & Koffler, D. (2022). Ifstat: Calculation of low flow statistics for daily stream flow data (version 0.9.12) [Computer software]. *CRAN*. Retrieved from https://cran.r-project.org/web/packages/lfstat/index.html
- Gustard, A., Bullock, A., & Dixon, J. M. (1992). Low flow estimation in the United Kingdom. Institute of Hydrology.
- Hailegeorgis, T. T., & Alfredsen, K. (2017). Regional flood frequency analysis and prediction in ungauged basins including estimation of major uncertainties for mid-Norway. Journal of Hydrology: Regional Studies, 9, 104–126. https://doi.org/10.1016/j.ejrh.2016.11.004
- Hamed, K. H. (2008). Trend detection in hydrologic data: The Mann–Kendall trend test under the scaling hypothesis. *Journal of Hydrology*, 349(3), 350–363. https://doi.org/10.1016/j.jhydrol.2007.11.009
- Hamon, W. R. (1963). Estimating potential evapotranspiration. Transactions of the American Society of Civil Engineers, 128(1), 324–338. https://doi.org/10.1061/TACEAT.0008673
- Han, S., Slater, L., Wilby, R. L., & Faulkner, D. (2022). Contribution of urbanisation to non-stationary river flow in the UK. Journal of Hydrology, 613, 128417. https://doi.org/10.1016/j.jhydrol.2022.128417
- Harman, C. J., Troch, P. A., & Sivapalan, M. (2011). Functional model of water balance variability at the catchment scale: 2. Elasticity of fast and slow runoff components to precipitation change in the continental United States. Water Resources Research, 47(2), 1–12. https://doi.org/10. 1029/2010WR009656
- Ho, M., Nathan, R., Wasko, C., Vogel, E., & Sharma, A. (2022). Projecting changes in flood event runoff coefficients under climate change. Journal of Hydrology, 615, 128689. https://doi.org/10.1016/j.jhydrol.2022.128689
- Hodgkins, G. A., Dudley, R. W., Archfield, S. A., & Renard, B. (2019). Effects of climate, regulation, and urbanization on historical flood trends in the United States. *Journal of Hydrology*, 573, 697–709. https://doi.org/10.1016/j.jhydrol.2019.03.102
- Horton, J. (1933). The Role of infiltration in the hydrologic cycle. *Eos, Transactions American Geophysical Union*, 14(1), 446–460. https://doi.org/10.1029/TR014i001p00446
- Ivancic, T. J., & Shaw, S. B. (2015). Examining why trends in very heavy precipitation should not be mistaken for trends in very high river discharge. *Climatic Change*, 133(4), 681–693. https://doi.org/10.1007/s10584-015-1476-1



- Jassby, A., Cloern, J., & Stachelek, J. (2022). wql: Exploring water quality monitoring data (version 1.0.0) [Computer software]. CRAN. Retrieved from https://cran.r-project.org/package=wql
- Kalbus, E., Reinstorf, F., & Schirmer, M. (2006). Measuring methods for groundwater—Surface water interactions: A review. *Hydrology and Earth System Sciences*, 10(6), 873–887. https://doi.org/10.5194/hess-10-873-2006

Kendall, M. G. (1948). Rank correlation methods. Griffin.

- Kirchner, J. W. (2024). Characterizing nonlinear, nonstationary, and heterogeneous hydrologic behavior using ensemble rainfall-runoff analysis (ERRA): Proof of concept. *Hydrology and Earth System Sciences*, 28(19), 4427–4454. https://doi.org/10.5194/hess-28-4427-2024
- Knoben, W. J. M., Woods, R. A., & Freer, J. E. (2018). A quantitative hydrological climate classification evaluated with independent streamflow data. Water Resources Research, 54(7), 5088–5109. https://doi.org/10.1029/2018WR022913
- Konapala, G., & Mishra, A. K. (2016). Three-parameter-based streamflow elasticity model: Application to MOPEX basins in the USA at annual and seasonal scales. *Hydrology and Earth System Sciences*, 20(6), 2545–2556. https://doi.org/10.5194/hess-20-2545-2016
- Lu, J., Sun, G., McNulty, S. G., & Amatya, D. M. (2007). A comparison of six potential evapotranspiration methods for regional use in the southeastern United States1. JAWRA Journal of the American Water Resources Association, 41(3), 621–633. https://doi.org/10.1111/j.1752-1688.2005.tb03759.x
- Mann, H. B. (1945). Nonparametric tests against trend. Econometrica, 13(3), 245-259. https://doi.org/10.2307/1907187
- McMillan, H. (2021). A review of hydrologic signatures and their applications. WIREs Water, 8(1), e1499. https://doi.org/10.1002/wat2.1499
  Neri, A., Villarini, G., Slater, L. J., & Napolitano, F. (2019). On the statistical attribution of the frequency of flood events across the U.S. Midwest. Advances in Water Resources, 127, 225–236. https://doi.org/10.1016/j.advwatres.2019.03.019
- Saft, M., Peel, M. C., Western, A. W., Perraud, J.-M., & Zhang, L. (2016). Bias in streamflow projections due to climate-induced shifts in catchment response. *Geophysical Research Letters*, 43(4), 1574–1581. https://doi.org/10.1002/2015GL067326
- Saft, M., Western, A. W., Zhang, L., Peel, M. C., & Potter, N. J. (2015). The influence of multiyear drought on the annual rainfall-runoff relationship: An Australian perspective. *Water Resources Research*, 51(4), 2444–2463. https://doi.org/10.1002/2014WR015348
- Sankarasubramanian, A., Vogel, R. M., & Limbrunner, J. F. (2001). Climate elasticity of streamflow in the United States. Water Resources Research, 37(6), 1771–1781. https://doi.org/10.1029/2000WR900330
- Schaake, J. C. (1990). From climate to flow. In P. E. Waggoner (Ed.), Climate change and US water resources (Vol. 8, pp. 177–206). Retrieved from https://api.semanticscholar.org/CorpusID:140694770
- Sharma, A., Wasko, C., & Lettenmaier, D. P. (2018). If precipitation extremes are increasing, why aren't floods? Water Resources Research, 54(11), 8545–8551. https://doi.org/10.1029/2018WR023749
- Sivapalan, M., Yaeger, M. A., Harman, C. J., Xu, X., & Troch, P. A. (2011). Functional model of water balance variability at the catchment scale: 1. Evidence of hydrologic similarity and space-time symmetry. *Water Resources Research*, 47(2). https://doi.org/10.1029/2010WR009568
- Slater, L. J., & Villarini, G. (2016). Recent trends in U.S. flood risk. Geophysical Research Letters, 43(24), 12428–12436. https://doi.org/10.1002/ 2016GL071199
- Slater, L. J., & Villarini, G. (2017). Evaluating the drivers of seasonal streamflow in the U.S. Midwest. Water, 9(9), 695. https://doi.org/10.3390/ w9090695
- Staudinger, M., Stoelzle, M., Seeger, S., Seibert, J., Weiler, M., & Stahl, K. (2017). Catchment water storage variation with elevation. *Hydrological Processes*, 31(11), 2000–2015. https://doi.org/10.1002/hyp.11158
- Steinschneider, S., Yang, Y.-C. E., & Brown, C. (2013). Panel regression techniques for identifying impacts of anthropogenic landscape change on hydrologic response. Water Resources Research, 49(12), 7874–7886. https://doi.org/10.1002/2013WR013818
- Stoelzle, M., Schuetz, T., Weiler, M., Stahl, K., & Tallaksen, L. M. (2020). Beyond binary baseflow separation: A delayed-flow index for multiple streamflow contributions. *Hydrology and Earth System Sciences*, 24(2), 849–867. https://doi.org/10.5194/hess-24-849-2020
- Tague, C., & Grant, G. E. (2009). Groundwater dynamics mediate low-flow response to global warming in snow-dominated alpine regions. Water Resources Research, 45(7). https://doi.org/10.1029/2008WR007179
- Tang, Y., Tang, Q., Wang, Z., Chiew, F. H. S., Zhang, X., & Xiao, H. (2019). Different precipitation elasticity of runoff for precipitation increase and decrease at watershed scale. *Journal of Geophysical Research: Atmospheres*, 124(22), 11932–11943. https://doi.org/10.1029/ 2018JD030129
- Tang, Y., Tang, Q., & Zhang, L. (2020). Derivation of interannual climate elasticity of streamflow. Water Resources Research, 56(11), e2020WR027703. https://doi.org/10.1029/2020WR027703
- Vicente-Serrano, S. M., Peña-Gallardo, M., Hannaford, J., Murphy, C., Lorenzo-Lacruz, J., Dominguez-Castro, F., et al. (2019). Climate, irrigation, and land-cover change explain streamflow trends in countries bordering the Northeast Atlantic. *Geophysical Research Letters*, 10821–10833. https://doi.org/10.1029/2019GL084084
- Vogel, R. M., Wilson, I., & Daly, C. (1999). Regional regression models of annual streamflow for the United States. *Journal of Irrigation and Drainage Engineering*, 125(3), 148–157. https://doi.org/10.1061/(ASCE)0733-9437(1999)125:3(148)
- Wasko, C., & Nathan, R. (2019). The local dependency of precipitation on historical changes in temperature. *Climatic Change*, 156(1), 105–120. https://doi.org/10.1007/s10584-019-02523-5
- Wasko, C., Nathan, R., & Peel, M. C. (2020). Changes in antecedent soil moisture modulate flood seasonality in a changing climate. Water Resources Research, 56(3), e2019WR026300. https://doi.org/10.1029/2019WR026300
- Zhang, Y., Zheng, H., Zhang, X., Leung, L. R., Liu, C., Zheng, C., et al. (2023). Future global streamflow declines are probably more severe than previously estimated. *Nature Water*, *1*(3), 1–11. https://doi.org/10.1038/s44221-023-00030-7

23284277, 2025, 7, Download